

PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (<http://bmjopen.bmj.com/site/about/resources/checklist.pdf>) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

TITLE (PROVISIONAL)	Examination of psychological risk factors for chronic pain following cardiac surgery: Protocol for a prospective observational study
AUTHORS	McGillion, Michael; Henry, Shaunattonie; Busse, Jason; Ouellette, Carley; Katz, Joel; Choinière, Manon; Lamy, Andre; Whitlock, Richard; Pettit, Shirley; Hare, Jacqueline; Gregus, Krysten; Brady, Katheryn; Dvirnik, Nazari; Yang, Stephen; Parlow, Joel; DuMerton-Shore, Deborah; Gilron, Ian; Buckley, D; Shanthanna, Harsha; Carroll, Sandra; Coyte, Peter; Ebrahim, Shanil; Isaranuwachai, Wanrudee; Guerriere, Denise N.; Hoch, Jeffrey; Khan, James; MacDermid, Joy; Martorella, Geraldine; Victor, J. Charles; Watt-Watson, Judy; Howard-Quijano, Kimberly; Mahajan, Aman; Chan, MTV; Clarke, Hance; Devereaux, P.J.

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.